 Vc/20-Apr-05
Week 16  starting 18 April 05
SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday  18 April 05 @ 4/18/2005 2:04:50 PM   
	11
	A1
	MBBR3316
	PT 13.1 IAP @ Warm
	294.46

	5
	A2
	MBAL2169
	PTE Min. NRJ 3KA Diode Test
	4.32 HV

	7
	B1
	SSS170
	PT 3 QH Measure
	2.31 HV

	2
	B2
	SSS130
	PT 12 Warm Up
	6.90

	10
	C1
	MBBL1111
	Prep 5 Cool Down
	279.92

	6
	C2
	MBBL1218
	Prep 5 Cool Down
	1.90

	1
	D1
	SSS151
	Stretched Wire Warm SW13
	292.67

	4
	D2
	SSS095
	PT 9 CDAP HV
	1.91

	8
	E1
	MBBL2164
	ICS 4 Final connection
	293.21

	3
	E2
	MBBL1216
	PT 6.2 Training 2
	1.90

	9
	F1
	MBAR3329
	Prep 5 Cool Down
	285.53

	12
	F2
	MBBR3242
	ICS 1 WP04 HV Test Warm
	292.05


Temperature data are taken from TBSM18 site
Coordinator this week: V Chohan; MTM support on call :   16-3602 .

19  complete Magnet Tests cycles   ( 3636 counted twice)
12  dipoles( 1 with MM)+ 6  SSS( 0  with MM) +  0  SSSS (with MM)
    Disconnected after tests last week ( Sat Midnight to Sat midnight) 

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· Situation on Repair :
General Points 
· SSS on D1 Next 

· New e-Traveller upgrades for Shafts
Tests 

	Magnets under Test from 10-04-2005 to 16-04-2005 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	#
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	

	1
	TBA1
	MBBL1213 
	[image: image1.png]



	


	09-04-2005 
	1 
	28 h.
	28 h.
	30 h.
	15 h.
	7 h.
	13-04-2005 
	108 h.
	Stripping
	

	2
	TBA2
	MBAL1205 
	[image: image3.png]



	


	10-04-2005 
	1 
	38 h.
	30 h.
	25 h.
	16 h.
	8 h.
	15-04-2005 
	117 h.
	Stripping
	

	3
	TBA2
	MBBR3307 
	[image: image5.png]



	


	04-04-2005 
	1 
	38 h.
	25 h.
	39 h.
	17 h.
	10 h.
	10-04-2005 
	129 h.
	Stripping
	

	4
	TBB1
	SSS132 
	[image: image7.png]



	


	11-04-2005 
	1 
	21 h.
	34 h.
	16 h.
	7 h.
	12 h.
	15-04-2005 
	90 h.
	Stand By
	

	5
	TBB1
	SSS133 
	[image: image9.png]



	


	08-04-2005 
	1 
	28 h.
	13 h.
	16 h.
	10 h.
	8 h.
	11-04-2005 
	75 h.
	Stripping
	

	6
	TBB2
	SSS010 
	[image: image11.png]



	


	10-04-2005 
	1 
	26 h.
	17 h.
	33 h.
	17 h.
	6 h.
	15-04-2005 
	99 h.
	Stripping
	

	7
	TBB2
	SSS092 
	[image: image13.png]



	


	05-04-2005 
	1 
	37 h.
	15 h.
	27 h.
	11 h.
	19 h.
	10-04-2005 
	109 h.
	Stripping
	

	8
	TBC1
	MBBL3636 
	[image: image15.png]



	


	07-04-2005 
	2 
	22 h.
	37 h.
	60 h.
	16 h.
	0 h.
	13-04-2005 
	135 h.
	Not Found
	

	9
	TBC1
	MBBL3636 
	[image: image17.png]



	


	13-04-2005 
	2 
	0 h.
	35 h.
	32 h.
	16 h.
	9 h.
	16-04-2005 
	92 h.
	Stand By
	

	10
	TBC2
	MBAL2168 
	[image: image19.png]



	


	10-04-2005 
	1 
	30 h.
	47 h.
	19 h.
	22 h.
	13 h.
	16-04-2005 
	131 h.
	Stand By
	

	11
	TBC2
	MBAR3320 
	[image: image21.png]



	


	05-04-2005 
	1 
	38 h.
	20 h.
	27 h.
	14 h.
	15 h.
	10-04-2005 
	114 h.
	Stand By
	

	12
	TBD1
	SSS102 
	[image: image23.png]



	


	07-04-2005 
	1 
	39 h.
	17 h.
	17 h.
	8 h.
	23 h.
	11-04-2005 
	104 h.
	Stripping
	

	13
	TBD2
	SSS095 
	[image: image25.png]



	


	07-04-2005 
	2 
	70 h.
	16 h.
	25 h.
	7 h.
	2 h.
	12-04-2005 
	120 h.
	Not Found
	

	14
	TBE1
	MBBR3318 
	[image: image27.png]



	


	08-04-2005 
	1 
	29 h.
	31 h.
	18 h.
	14 h.
	6 h.
	12-04-2005 
	98 h.
	Stripping
	

	15
	TBE2
	MBAL1208 
	[image: image29.png]



	


	10-04-2005 
	1 
	23 h.
	30 h.
	24 h.
	17 h.
	5 h.
	14-04-2005 
	99 h.
	Stripping
	

	16
	TBE2
	MBBL2160 
	[image: image31.png]



	


	06-04-2005 
	1 
	22 h.
	26 h.
	20 h.
	16 h.
	12 h.
	10-04-2005 
	96 h.
	Stripping
	

	17
	TBF1
	MBAR3303 
	[image: image33.png]



	


	12-04-2005 
	2 
	24 h.
	34 h.
	21 h.
	19 h.
	9 h.
	16-04-2005 
	107 h.
	Stripping
	

	18
	TBF1
	MBBR3246 
	[image: image35.png]



	


	08-04-2005 
	1 
	25 h.
	24 h.
	23 h.
	16 h.
	11 h.
	12-04-2005 
	99 h.
	Stripping
	

	19
	TBF2
	MBAL2153 
	[image: image37.png]



	


	09-04-2005 
	1 
	34 h.
	26 h.
	17 h.
	17 h.
	5 h.
	13-04-2005 
	99 h.
	Stripping
	


Quench Information
	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBL1213
	12475 A.
(D1_L)
	12626.5 A.
(D2-L)
	12694 A.
(D2-L)
	[image: image39.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBAL1205
	11863 A.
(D1_L)
	12520 A.
(D1 - U)
	12630 A.
(D1 - U)
	[image: image40.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBBR3307
	12230.7 A.
(D2 - U)
	12680 A.
(D2-L)
	12848 A.
(D2_U)
	[image: image41.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	SSS132
	12233.7 A.
(QI_12)
	12396.4 A.
(QE-34)
	12842 A.
(QI_12)
	[image: image42.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	SSS133
	12340 A.
(QI_12)
	12340 A.
(QE_34)
	12458 A.
(QE-34)
	[image: image43.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	SSS010
	11684 A.
(QI-34)
	12067 A.
(QI-12)
	12850 A.
[image: image44.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS092
	11588 A.
(QI-12)
	11782 A.
(QI-34)
	11796.3 A.
(QI_34)
	12196 A.
(QE-12)
	12326 A.
(QI-34)
	12850 A.
[image: image45.jpg]



	-
	-
	-
	-
	-
	-
	-

	8
	MBBL3636
	9129 A.
(D2_U)
	11759 A.
(D2_U)
	11729 A.
(D2_U)
	12004 A.
(D1-U)
	12065 A.
(D2_U)
	12371.7 A.
(D2-U)
	12651 A.
(D2-U)
	12800 A.
(D2-U)
	12850 A.
[image: image46.jpg]



	-
	-
	-
	-

	9
	MBBL3636
	11011 A.
(D1-L)
	12045 A.
(D2_L)
	11980 A.
(D2-L)
	0 A.
(No Data)
	12288 A.
(D2_U)
	[image: image47.png]



	-
	-
	-
	-
	-
	-
	-

	10
	MBAL2168
	12019 A.
(D1-U)
	12617 A.
(D2_L)
	12508 A.
(D1-U)
	[image: image48.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBAR3320
	10849 A.
(D2-L)
	12644.8 A.
(D1-U)
	12734 A.
(D1_L)
	[image: image49.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	SSS102
	11900 A.
(QI - 12)
	12429 A.
(QI-12)
	12540 A.
(QI-34)
	[image: image50.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	13
	SSS095
	12192.5 A.
(QE_34)
	0 A.
(No Data)
	11714 A.
(QI-12)
	11897 A.
(QE34)
	12791 A.
(QE34)
	12850 A.
[image: image51.jpg]



	-
	-
	-
	-
	-
	-
	-

	14
	MBBR3318
	12757.2 A.
(D1_L)
	12850 A.
[image: image52.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	15
	MBAL1208
	12720.3 A.
(D1-L)
	11858 A.
(D1_L)
	12632.9 A.
(D1-L-S01)
	12856 A.
[image: image53.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	16
	MBBL2160
	11270.7 A.
(D2 - U)
	11834.1 A.
(D1-U)
	12258.5 A.
(D2-U)
	12444 A.
(D1-L)
	[image: image54.png]



	-
	-
	-
	-
	-
	-
	-
	-

	17
	MBAR3303
	11923 A.
(D2-U)
	12253 A.
(D2-U)
	12572 A.
(D2-U)
	[image: image55.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	18
	MBBR3246
	12450 A.
(D2-U-S01)
	12549 A.
(D2_L_S01)
	12850 A.
[image: image56.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	19
	MBAL2153
	11451 A.
(D1-L)
	12087 A.
(D2 - U)
	12513 A.
(D2-L)
	[image: image57.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-


Week 15  :  10 April to 16 April 05  (19 Magnet Tests)
[   3636 On bench rapid Thermal cycle   ]
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	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday _12_N

Thursday _14_N

Monday_18_A


	TBE2

TBE1

TBC2
	HV test - cold initial. Magnet was at 1.9K, but checks by the PLC of the HV rack (signal taken from the PLCSE) are indicating that the phase is not stable and impede to do the measurement.. 
HV at cold not able to run PLC error " Wrong cryo phase", though HV-OK signal is available according to Cryo operators.
	ACR

ACR

ACR
	
	Guy called and disabled the PLC check.

Advise of Richard :In case of problems during HV tests, related with the cryo phase, ask the cryo OP team  will look to the trends to see if the signal "Cold_HV signal" is stable. If not wait until it becomes stable.
Guy, suggested to wait for more time the stability of the phase.

Impossible to restart the test. Richard investigated with ACR the problem.  Bug in the ACR software  when two benches of the same cluster were in HCV conditions.?
Same problem

	Tuesday _12_N

Wednesday _13_N


	TBB2
	Rack XMTY92: BNC connector of cable nr 16 for the passive bus terminal need to be repaired. 
Only 2 HV racks were present, we could not locate 3rd Rack, so HV test PT9 was not launched
	MTM
	
	Richard removed the rack for repair. Repaired Thursday morning.

Only 2 racks available Wednesday. 3 tests at the same time cannot be done.



	Wednesday _13_M


	TBA2
	Faulty quench at 40A on MBB1205
	MTM
	
	

	Wednesday _13_N


	TBC2
	Cryo OK (HV) lost during HV test (2 times)
	ACR
	
	Test aborted.

	Thursday _14_M


	TBC2
	Noise apparent on signals for training quench  1 on C2 MBAL2168  alarm: D1-2 Dirvt direct voltage jump exceeded 10V
	MTM
	
	File checked by Guy: OK to continue.

	Thursday _14_M


	TBB1
	Resistance of one quench heater of the SSS132 (those of the dummy resistors) is not correct. 
	MTM
	
	Guy  found that he discharge damaged the dummy resistor and its power supply. He changed the faulty power supply and the dummy resistors.

	Thursday _14_A

Thursday _14_N

Fridday_15_M

Saturday_16_A


	TBD1
	SPA at 5000 A and at 11850 A.

SPA during training quench
	ACR?
	
	Problem in the  level gage in the exchanger : peak of He level in the exchanger  detected.

	Thursday _14_N

Friday_15_M


	TBD1
	Wire broken during FS measurement at 760 A.

Idem

Config. Files related to the longitudinal position of correctors are also missing.
	MTM
	
	Test repeated.

Guy inserted the files

	Friday _15_N


	TBF1
	Magnet names on F1 and F2 are not shown on TEMA screen.  Communication error “No info from UDP socket”.
	AB/CO
	
	Guy switched off/on the sun.

	Friday _15_N


	TBD1
	XY axis measurements in aperture 2 with the SSW system n3 give bas results (DC measurements).
	AB/CO
	
	OP team called Guy who came and decided to change the measurements system. The system n2 was installed.

	Saturday _16_A


	TBF2
	Not possible to launch TEMA on the server. Communication problem from the VME crate
	AB/CO
	
	Guy reset the crates and OK.

	Saturday _16_N


	TBD1
	Shaft not rotating correctly (sliding) in aperture 2
	
	
	Guy came and repaired the clutch.

	Sunday _17_A


	TBB1
	EF Corrector measurements on SSS170 not possible,
	ICS
	
	Guy suspected an inversion of voltage taps. He fixed the problem by disabling the Potaim cards of these Vtaps.

	Monday_18_M


	TBB2
	SPA during training quench
	ACR
	
	Too low He gaz flow for the regulation  due to a too low loss in these CFB, unable maintain the current leads at correct temperature.

	Monday_18_M


	TBA1
	IAP@warm final on MB3316: error in D1,D2  values.
	MTM
	
	Federico checked. Faulty potaim card. Magnet OK.

	Monday_18_M


	TBD2
	Potential has been received during voltage taps disconnection (switch in CC position). 

To be clarified.???
	MTM
	
	

	Monday_18_A


	TBE2
	HF DAQ could not be armed.
	AB/CO
	
	Exit from all applications & started Tema after reseting the crates, problem solved.











